Author's response to reviews

Title: Usefulness of Multimodal MR Imaging in the Differential Diagnosis of HaNDL and Acute Ischemic Stroke

Authors:

Tomas Segura (tsmtsm08@yahoo.es)
Francisco Hernandez-Fernandez (pachdezfdez@hotmail.com)
Pedro Sanchez-Ayaso (sanchezayaso@gmail.com)
Elena Lozano (elenalozano@ono.com)
Lorenzo Abad (lorenzoabad@ono.com)

Version: 8 Date: 29 November 2010

Author's response to reviews: see over
Dear sir:

We submit a new revised version of the manuscript entitled “Usefulness of Multimodal MR Imaging in the Differential Diagnosis of HaNDL and Acute Ischemic Stroke”.

These are the changes made:

- Each affiliation has been written in full in the suggested format
- The abstract has been reformatted
- The manuscript structure has been reformatted according to the proposed headings; given that consent section was not mentioned, it has not been included
- The table included in figure-1 legend has been included within the image itself
- Now we use sentence case in the title
- All text is in black
- We have removed key words
- We have removed additional files

We look forward to hearing from you. Thank you for your attention to our submission.

Yours faithfully,

Dr. Tomás Segura